Journal of Eating Disorders



The relationship between self-control and symptoms of anxiety and depression in patients with eating disorders: a cross-sectional study including exploratory longitudinal data

Emmi Eriksson¹, Mia Ramklint¹, Martina Wolf-Arehult^{1,2} and Martina Isaksson^{1*}

Abstract

Background Personality style can partly be described as the way an individual controls and regulates emotions and can be divided into over- and undercontrol. Studies have indicated that personality style may impact the onset, clinical presentation, and recovery from an eating disorder (ED). Furthermore, symptoms of anxiety and depression are common in patients with EDs. However, the association between self-control levels and anxiety/depression symptoms in patients with EDs remains unknown. The main aim of this study was to assess how levels of self-control relate to anxiety/depression symptoms in patients with EDs, with a secondary, exploratory aim to assess the stability of self-control during treatment.

Methods Patients were recruited from the outpatient ED clinic at the Uppsala University Hospital, between October 2014 and December 2019. In total, 227 patients (age: 25.4, SD: 7.1) were included at the start of their treatment, with 14 participants also completing post-treatment measurements. Self-control was assessed with the Ego Undercontrol scale (EUC-13), anxiety/depression symptoms with the Hopkins Symptoms Checklist (HSCL-25), and ED diagnosis and symptoms with the Eating Disorder Examination Interview (EDE-I) and Questionnaire (EDE-Q), respectively.

Results A quadratic regression (n = 227) showed that levels of self-control accounted for about four percent of the variance in degree of global anxiety/depressive symptoms. Anxiety/depression symptoms were better explained by ED symptoms ($R^2 = 0.24$). Visualizations in boxplots revealed a tendency for extreme values of both over- and undercontrol to be associated with higher levels of depression, whereas symptoms of anxiety increased with increasing undercontrol. In the exploratory analyses (n = 14) levels of self-control remained more stable than symptoms of anxiety and depression, which decreased significantly during ED treatment.

Conclusions Our results indicated that anxiety/depression symptoms, in patients with EDs, were not strongly correlated with levels of self-control, but rather with ED symptoms. However, extreme values of both over- and undercontrol showed a tendency to be associated with higher levels of depression symptoms, whereas anxiety symptoms

*Correspondence: Martina Isaksson martina.isaksson@neuro.uu.se Full list of author information is available at the end of the article



© The Author(s) 2023. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

increased with increasing levels of undercontrol. Future studies could benefit from considering both over- and undercontrol as potentially dysfunctional.

Keywords Eating disorders, Personality style, Anxiety, Depression, Self-control, Overcontrol, Undercontrol

Plain English summary

Eating disorders (EDs) are mental disorders characterized by a persevering behavioral disturbance in eating and distress in thoughts or emotions, significantly impairing psychosocial function and/or physical health. Anxiety and depression are common in patients with EDs.

Studies have indicated that personality style may impact the onset, clinical presentation, and recovery from an ED. Personality style can be studied in terms of self-control, ranging from over- to undercontrol. However, it remains unknown how self-control relates to anxiety/depression in patients with EDs and its stability during ED treatment. This is what we aimed to investigate in this study, collecting data before treatment in 227 adult patients from a Swedish outpatient ED clinic, with 14 also completing post-treatment measurements.

Our results indicated that global anxiety/depression levels in patients with EDs were not strongly correlated with levels of self-control, but rather with ED symptoms. Nevertheless, anxiety seemed to increase with increasing levels of undercontrol and extreme values of both over- and undercontrol showed a tendency to be associated with higher levels of depression. In the exploratory analyses self-control levels remained more stable during treatment than global anxiety/depression levels, which decreased significantly.

Background

Eating disorders (ED) are mental disorders characterized by a persevering eating behavioral disturbance and associated distress in thoughts or emotions, significantly impairing psychosocial function and/or physical health [1]. Symptoms of anxiety and depression are common in patients with EDs [2]. Some studies indicate that up to 65% of the patients presenting with an ED also meet the criteria for an anxiety disorder [2], and depression affecting 46-74% of patients with anorexia nervosa (AN) and 30–60% of those with bulimia nervosa (BN) [3]. The expected lifetime incidence of an affective disorder in patients with EDs is up to 98% [4]. EDs are often severe disorders with an elevated mortality rate ratio of 2.87 [5], remission rates of approximately 45% after treatment, i.e., remaining symptoms for the majority of patients, and a relapse rate of 30% within the first year among the individuals in remission [6]. Thus, there is a need for improved understanding and treatment of these disorders.

Individual variations are often overlooked in patients with eating disorders, which presents challenges in establishing an acceptable, effective and feasible first-line outpatient treatment [7]. A personality-oriented approach has been suggested as an alternative to established treatment or as a complementary way of optimizing treatment and predicting the clinical course and outcome in patients with EDs [7]. Several studies have identified three main personality styles among ED patients: an undercontrolled, an overcontrolled and a low psychopathology style, the latter also known as high-functioning perfectionistic or resilient [8–10]. These personality styles are related to the concept ego control, also known as self-control. Self-control refers to the ability to inhibit or express impulses to achieve long-term goals and can vary along a spectrum from undercontrol to overcontrol [11]. Undercontrol is characterized by impulsiveness, spontaneity, fluctuating emotions, and expressing these emotions even when socially inappropriate [11]. Overcontrol, on the other hand, can lead to overly organized behavior, as well as an ability to delay gratification and refrain from pleasure [11].

Self-control has previously been linked to ED treatment outcome, with undercontrolled patients having a higher risk of poor outcome at discharge [10]. However, it has also been shown that patients with an ED in addition to obsessive-compulsive personality disorder (OCPD) or autism spectrum disorder (ASD), diagnoses characterized by high levels of self-control or overcontrol, have poorer outcome compared with non-comorbid groups [12], hence suggesting that both overcontrol and undercontrol can be negative predictors for treatment outcome. Both over- and undercontrol have also been associated with more severe symptoms of mental disorders [13], including more severe ED symptoms [10, 14] compared to intermediate levels of self-control. Further, high levels of self-control are positively correlated with internalizing problems, such as depression and anxiety, whereas high levels of undercontrol have a positive correlation with externalizing problems [15], such as substance use disorder. Such externalizing problems are often associated with symptoms of anxiety and depression

as secondary effects [16, 17]. However, the relationship between self-control and degree of anxiety and depression symptoms in the ED population has not been investigated. Understanding whether high and/or low levels of self-control are related to symptom severity of anxiety and depression symptoms could improve future possibilities of individualizing ED treatments, targeting not only ED symptoms but also anxiety and depression symptoms and personality style.

Personality is, by definition, a pattern that is stable over time, in both non-clinical and clinical groups [18]. However, there is increasing evidence that maladaptive personality characteristics may be treatable [19] and expanded knowledge about the stability of self-control levels might thus not only improve our understanding of the etiology and presentation of EDs, but also aid in developing more efficient treatments.

The main aim of this study was to evaluate whether levels of self-control relate to symptoms of anxiety and depression in patients with EDs, with a secondary, exploratory aim to study the stability of self-control levels during ED treatment. Considering the known association between over- and undercontrol to internalizing and externalizing problems, respectively, we hypothesized that ED patients with a higher degree of either over- or undercontrol would have more severe symptoms of anxiety and depression than patients with intermediate ratings. The secondary aim was purely exploratory.

Methods

Procedure and participants

Patients were recruited from the ED clinic at Uppsala University Hospital between October 2014 and December 2019. The ED clinic is an outpatient clinic for patients with AN and atypical anorexia nervosa (AAN), moderate to severe BN, and other complex ED cases with comorbidities or suicidality. Binge eating disorder (BED), avoidant restrictive food intake disorder (ARFID), and subclinical BN are generally not treated at the clinic, but generally referred to the obesity unit, primary or private care or the general psychiatry. All patients referred to the clinic, fulfilling the inclusion criteria of a minimum age of 18 years and an ED diagnosis were eligible for this study, and in total 227 patients were included at the start of their treatment. The inclusion was broad to represent a clinical sample. The exclusion criteria included being assessed as unable to participate and fill out questionnaires due to severe mental illness, i.e., when patients needed inpatient care or other emergency interventions, had insufficient knowledge of Swedish, or had limited cognitive abilities. ED diagnoses were determined by any one of 15 psychologists, trained in the Eating Disorder Examination Interview (EDE-I) by one expert. To assess the inter-rater reliability six filmed interviews conducted by the expert, were assessed by all the 15 trained psychologists, who showed complete agreement in all ratings. As the University hospital, where the study was performed, used DSM-IV for diagnostics when the study was initiated, a decision was made to continue diagnosing according to the DSM-IV throughout. Afterwards, a recoding in accordance with the DSM-5 criteria was performed so that the sample could be presented in accordance with the revised diagnostic criteria. The group of other eating disorders included ARFID (n=2), other specified eating disorders (OSFED): BN of low frequency (n = 16), purging disorder (n = 14), and others (n = 21), where the reason for not fulfilling a specific ED diagnosis was another than exemplified OSFED groups in the DSM-5. BED (n=6) was also included in the group of other eating disorders, since it was rarely treated at the clinic.

All patients who decided to participate in the study received verbal and written information about the study at their first meeting during the psychological assessment phase and gave written informed consent. Weight and height measurements were performed at the clinic and background variables such as age, gender, marital status, occupation and level of education were collected at baseline.

At the time of their last therapy session, a subset of patients were asked to fill out questionnaires yet again. Twenty agreed, with 14 patients completing all questionnaires, i.e., EDE-Q, HSCL-25 and EUC-13 at both timepoints. Due to organizational issues at the clinic, the post-treatment measurements were not always handed out and collected as planned, explaining the small sample. The 14 patients had received Enhanced Cognitive Behavioral Therapy, typically consisting of 20 sessions for BN and 40 sessions for AN patients, over the same number of weeks. 10 had reached ratings within one SD of the mean of a non-clinical sample, indicating possible remission [Eating Disorder Examination Questionnaire (EDE-Q) ≤ 2.83]. The recruitment process is shown in Fig. 1.

Instruments

EDE-I and EDE-Q

All participants underwent the semi-structured EDE-I at baseline, to determine ED diagnosis. The interview was conducted by a trained psychologist and focused on attitudes and behavior central to ED pathology [20].

The EDE-Q is a well-established self-report instrument with 36 items, derived from the EDE-I, to assess ED symptoms. The attitudinal items are divided into four subscales—restraint, eating concern, shape concern, and weight concern—and are rated by patients for severity or frequency, using a 7-point rating scale. Responses to behavioral questions, for instance on binge eating and



Fig. 1 Flowchart of participant recruitment. AAN = Atypical Anorexia Nervosa. AN = Anorexia Nervosa. BN = Bulimia Nervosa. ED = Eating Disorder

vomit induction, are given as number of days. The level of ED psychopathology increases with a higher score. As with the EDE-I, the questions cover the preceding 28 days. Swedish norms are available with a mean score of 1.56 (SD=1.27), with 84% of the population scoring 2.83 or below, in the non-clinical population [21]. Considering the prevalence of EDs, scores below this value can be used as an indication of EDE-Q values similar to

those in a non-clinical population and, in the absence of other measures, as a proxy for remission or at least partial remission. Correlations with other measures of similar constructs range from 0.40 to 0.79 for the different subscales [22].

Data support the test–retest reliability of both EDE-I and EDE-Q, as well as their validity in differentiating those with ED pathology from non-clinical controls. The internal consistency of EDE-Q is acceptable to high, with the different subscales ranging from coefficient alpha 0.7 to 0.93 [22].

EUC-13

The Ego Undercontrol Scale-short version (EUC-13) is a self-rating questionnaire that contains 13 items, measuring levels of expression and inhibition of emotional impulses on a single spectrum ranging from over- to undercontrol. The EUC-13 is based on a three-factor model consisting of the subscales uninhibited behavior (five items), planful conscientious behavior (five items), and socially restrained behavior (three items). The items on each subscale are rated on a four-point Likert scale, ranging from "Disagree very strongly" [1] to "Agree very strongly" [4]. When calculating the final score, the subscales of planful conscientious behavior and socially restrained behavior are reversed. Thus, a high score indicates low levels of emotional impulse inhibition (high undercontrol). The EUC-13 has been validated in a nonclinical Swedish community sample, showing adequate test-retest reliability, in addition to construct validity and acceptable internal consistency ($\alpha = 0.71$) [23]. However, use of the global score is recommended due to its higher validity [24].

HSCL-25

The Hopkins Symptom Checklist-25 (HSCL-25) is a 25-item self-report questionnaire designed to screen for anxiety (10 items) and depression (15 items) symptoms. The respondent rates each symptom, depending on how bothersome it has been during the week prior to assessment, on a scale ranging from "Not at all" [1] to "Extremely" [4, 25]. The scale has been widely used for screening, with a cut-off of 1.75, demonstrating accuracy when validated against various diagnostic interviews worldwide and has a high internal consistency, with a coefficient alpha of 0.80 for both subscales, respectively [26]. The scale's median average score has also shown high correlation with other measures of mental health [27], and satisfactory validity and reliability in assessing mental symptoms [28]. The scale is validated in a Swedish sample [26].

Data collection and analyses

Missing values for single questions in EUC-13 and HSCL-25 were handled through single imputation using the mean of the affected subscale, as long as the missing values were limited to less than 20% of the subscale, in accordance with recommendations [29]. If this limit was exceeded, the entire scale was omitted for that individual. However, missing values were rare (<0.5%). For the EDE-Q, a global score was calculated as the mean score of all rated items, as long as at least 50% of the items were rated, in accordance with the scoring recommendations [30].

Normal distribution of the data was checked by visualization in histograms, and through the values of skewness and kurtosis, and data were further checked for outliers using box- and scatterplots.

The relationship between global HSCL-25 scores and the EUC-13 was initially visualized in a scatterplot and through locally estimated scatterplot smoothing (LOESS), which indicated a slight curvilinear trend. An analysis of variance (ANOVA) was conducted, further strengthening indications of a quadratic term, with a higher R^2 and a lower standard error of the estimate, in addition to a significant F-ratio and nonsignificant Levene's test. Additionally, a linear regression was tested, resulting in similar, but somewhat lower, explained variance than in the quadratic regression. A quadratic regression model was therefore chosen, with global HSCL-25 score as the dependent variable, as well as a model with the subscale for depressive symptoms as the dependent variable. As the LOESS of levels of self-control in relation to symptoms of anxiety showed a linear trend, a linear regression was chosen for this analysis. Assumptions for these models were tested, all of which showed the data as suitable for regression models. The analysis was also conducted excluding the rarer and atypical eating disorders OSFED: ARFID, purging disorder, and OSFED: others with essentially unchanged results, and these participants were therefore included in all analyses.

Next, a visual representation of potential confounders was performed in a directed acyclic graph. Adjustments for gender, age, and starvation were found to be necessary to estimate the effect of EUC-13 scores on HSCL-25 scoring, for which we applied BMI as a proxy. The potential confounding effect of EDE-Q was more unclear, depending on whether the relationship between EUC-13 and EDE-Q scoring was considered uni- or bidirectional. Subsequently, two adjusted models were created, one correcting only for gender, age, and BMI, and the other also including EDE-Q scores. Correlations between the described factors and the outcome, HSCL-25, were tested, and the factors were checked for data multicollinearity prior to analyses, showing weak to non-significant correlations. Data on confounding factors were available for 220 of the participants.

To examine how extreme values of self-control, i.e., over- and undercontrol, related to the degree of symptoms of anxiety and depression, measured with HSCL-25, the participants were divided into four groups, based on their EUC-13 scores, for a subset of analyses. These groups were defined by standard deviation (SD) from the sample mean, in steps of 1 SD. The number of groups was chosen to enable comparison between what may be considered functional levels of over- and undercontrol (in the groups -1 and +1 SD from the sample mean, respectively) and those with more extreme levels, while still ensuring sufficient numbers of participants in each group. This generated the following groups: highly overcontrolled (n=32), overcontrolled (n=101), undercontrolled (n=68), and highly undercontrolled (n=26). The analyses conducted in these groups included visualization in boxplots for the two subscales anxiety and depression symptoms, a one-way ANOVA, and Tukey's test.

For the exploratory analyses in the subgroup of participants with both pre- and post-treatment measurements, sample differences were tested with Wilcoxon's signedrank test, which was chosen due to the small sample size. Data was further visualized in side-by-side boxplots and the reliable change index was calculated for each individual.

Analyses were all conducted with SPSS Statistics version 28.0.1.0.

Results

Participant demographics are presented in Table 1.

EUC-13 and HSCL-25

The quadratic regression showed a statistically significant trend that patients exhibiting over- and undercontrol on the EUC-13 tended to score higher on global anxiety/ depression symptoms, measured using HSCL-25 [F (2, (224) = 4.98, p = 0.008]. The EUC-13 was a statistically significant coefficient (p = 0.009) in the model. However, the proportion of variance explained by the EUC-13 was low ($R^2 = 0.043$): four percent of the variance in anxiety/ depression symptoms was explained by self-control. The graph for this is presented in Fig. 2. When running the same analysis for the subscale covering depression, and a linear regression for anxiety symptoms, the explained variance was essentially the same $(R^2 = 0.03$ for both depression and anxiety symptoms), with both over- and undercontrol being associated with depression and anxiety symptoms increasing with increasing levels of undercontrol.

Table 1 Descriptive data on participants (n = 227) at baseline

Gender	
Woman	216 (95.2%)
Man	9 (4.0%)
Other	2 (0.9%)
Mean age, years (SD) (Range)	25.4 (7.1) (18–59)
Mean BMI, kg/m ² (SD) ^a (Range)	21.4 (4.8) (15.0–40.4)
DSM 5 diagnosis	
AN	66 (29%)
AAN	27 (11.9%)
BN	75 (33.0%)
Other ED	59 (25.9%)
Level of education ^b	
Elementary school	30 (13.2%)
High school	113 (49.8%)
Higher education	82 (36.1%)
Main occupation ^c	
Paid work	78 (34.4%)
Student	102 (44.9%)
Sick leave	38 (16.7%)
Unemployed	3 (1.3%)
Other	4 (1.8%)
Marital status ^d	
Single	131 (57.7%)
Married/Relationship	79 (34.8%)
Other	15 (6.6%)

AAN Atypical Anorexia Nervosa, AN Anorexia Nervosa, BN Bulimia Nervosa, ED Eating Disorder, SD Standard deviation

^a 3 patients missing data on BMI

^b 1 missing level of education

^c 2 missing main occupation

^d 2 missing marital status

The results remained stable, and EUC-13 remained a statistically significant coefficient (p=0.034) in the model, when adjustments were made for gender, age and BMI ($R^2=0.078$, p=0.007). However, when the analysis was conducted controlling for gender, age, BMI, and EDE-Q scores, the effect of the EUC-13 scores was non-significant (p=0.09), with EDE-Q scores being the strongest predictor of global levels of anxiety/depression symptoms ($R^2=0.24$, p<0.001).

Visualizations of the HSCL-25 subscales anxiety and depressive symptoms in boxplots for the four EUC-13 groups are presented in Fig. 3. In the ANOVA, the difference between groups for the subscale covering anxiety symptoms was statistically significant (p=0.040, $\eta 2=0.036$) though that for depressive symptoms was not (p=0.30, $\eta 2=0.016$). A follow-up Tukey's test showed



Fig. 2 Scatterplot and quadratic model of global anxiety/depression symptoms in relation to self-control levels. Anxiety/depression symptoms are measured with Hopkins symptom checklist 25 (HSCL-25) and levels of self-control with Ego Undercontrol scale 13 (EUC-13) in a group of patients with an eating disorder diagnosis (n = 227), pre-treatment

that the difference in anxiety levels between groups stemmed from the difference between the highly overand highly undercontrolled groups (p=0.044). The distribution between different ED diagnosis in the different groups of self-control levels, based on EUC-13, is presented in Table 2.

EUC-13 and HSCL-25 pre- and post-ED treatment: exploratory analyses

The mean pre-treatment EUC-13 value was 2.36 (SD = 0.60), decreasing to 2.21 (0.40) post-treatment (n = 14). A Wilcoxon's signed-rank test showed that this difference in the mean was non-significant (p = 0.17). However, the within-group variance decreased from 0.36 pre-treatment to 0.16 post treatment, and from 0.40 to 0.09 in the subgroup of patients who had achieved possible remission (n=10). Among these nine cases, four showed reliable changes, all towards the mean, whereas five remained unchanged or with non-reliable changes. For HSCL-25, the global mean pre-treatment score was 2.67 (SD = 0.69), with a significant decrease to 1.96 (0.51) post-treatment (Wilcoxon, p = 0.004). Individual mean differences pre- and post-treatment in both EUC-13 and HSCL-25, and the calculated reliable change index (RCI) are presented in Table 3.

Discussion

The main aim of this study was to evaluate whether levels of self-control relate to symptoms of anxiety and depression in patients with EDs, with a secondary, exploratory, aim to study the stability of self-control levels during ED treatment.

The main finding was that the levels of anxiety and depression symptoms in patients with EDs were not strongly correlated with levels of self-control, but rather with ED symptoms. However, there was a tendency for extreme values of both over- and undercontrol to be associated with higher global levels of anxiety/depression symptoms, supporting our initial hypothesis. When the subscales were studied separately, symptoms of anxiety seemed to increase with increasing levels of undercontrol, whereas depressive symptoms were somewhatthough not statistically significantly-higher in both the most over- and the most undercontrolled. In our exploratory analyses, self-control levels seemed to remain more stable at the post-treatment assessment than the level of symptoms of anxiety and depression, which decreased significantly.

The proportion of global anxiety/depression symptoms explained by levels of self-control was low (about 4%), but statistically significant. These results were stable when investigating the impact of the confounding factors of gender, age, and BMI. However, when correcting for



Fig. 3 Boxplots of symptoms of depression (top) and anxiety symptoms (bottom) in relation to self-control levels. Symptoms of depression and anxiety are assessed with Hopkins symptom checklist 25 (HSCL-25) and the degree of self-control is measured with Ego undercontrol scale 13 (EUC-13), in a group of patients with an eating disorder, pre treatment. The results are presented in four groups with different levels of self-control, based on the standard deviation from the sample EUC-13 mean score (n = 227): highly overcontrolled (n = 32), overcontrolled (n = 101), undercontrolled (n = 68), and highly undercontrolled (n = 26)

EDE-Q scores, the impact of self-control levels lost significance. It is therefore possible that other factors, such as the degree of ED symptoms, are of greater importance than personality style for predicting levels of anxiety and depression symptoms. Nevertheless, the impact of the EDE-Q as a confounder is questionable, as extremely high or low levels of self-control may increase ED symptoms, which in turn would increase levels of anxiety and depression symptoms. Thus, ED symptoms may form part of the causal pathway of self-control generating higher global levels of anxiety/depression symptoms; adjusting for the EDE-Q may therefore lead to overcorrection [31]. It can be argued that the model adjusted only for gender, age, and BMI was the most appropriate one. Additionally, the low levels of explained variance in our model may, at least partly, be explained by the scarcity of participants exhibiting pronounced levels of overand undercontrol, giving them less impact in the model.

Diagnosis	Highly overcontrolled < (- 1 SD)	Overcontrolled (1 SD) Mean	Undercontrolled Mean – (+ 1 SD)	Highly undercontrolled > + 1 SD
AN	13 (40.6%)	34 (33.7%)	15 (22%)	4 (15.4%)
AAN	5 (15.6%)	16 (15.8%)	5 (7.3%)	1 (3.8%)
BN	8 (25%)	23 (22.8%)	27 (39.7%)	17 (65.4%)
Other ED	6 (18.8%)	28 (27.78%)	21 (30.9%)	4 (15.4%)

Table 2 Distribution of eating disorder diagnosis in different groups of self-control levels

Self-control levels are measured with Ego undercontrol scale 13 (EUC-13), in a group of patients with an eating disorder, pre-treatment. The groups are based on the standard deviation from the sample EUC-13 mean score (n = 227) and consist of highly overcontrolled (n = 32), overcontrolled (n = 101), undercontrolled (n = 68), and highly undercontrolled (n = 26)

AAN Atypical Anorexia Nervosa, AN Anorexia Nervosa, BN Bulimia Nervosa, ED Eating Disorder

When the subscales anxiety and depressive symptoms were studied separately, the degree of anxiety symptoms seemed to increase with increasing levels of undercontrol, whereas levels of depression symptoms showed tendencies of a curvilinear trend with both over- and undercontrol being associated with higher levels of depressive symptoms. However, the effect sizes between groups of different levels of self-control were small.

Our findings add some evidence to the conceptualization of self-control as a spectrum, with both strong under- and overcontrol being associated with higher degrees of internalizing symptoms such as anxiety/ depression symptoms. This is in contrast to the views of some theorists who consider only undercontrol to be disadvantageous [10]. Instead, our findings are in line with earlier studies associating both high levels of over- and undercontrol with more severe symptoms in patients with mental disorders [13]. However, the fact that the relationship between the subscales anxiety and depressive symptoms and self-control levels differ may indicate that the mechanism behind these symptoms varies between the overcontrolled and undercontrolled groups. Previous studies have linked overcontrol to social isolation [32, 33] and it has been suggested that depressive symptoms could be a consequence of such isolation. Undercontrol, on the other hand, is associated with impulsiveness, aggression, and more frequent conflicts [34], which may be crucial factors generating higher levels of anxiety and depression symptoms. The idea that the personality styles may cause different behavioral, emotional, and attitudinal problems raises the question of whether it could be beneficial to individualize therapy based on levels of self-control.

Further, differences in the distribution between type of ED diagnosis could be seen in the groups of different self-control levels. AN predominated in the highly over-controlled group, and BN in the highly undercontrolled group, in line with previous studies [35, 36].

In the exploratory measurements before and after ED treatments, the EUC-13 mean value remained stable, and the majority of the studied individuals showed no significant tendency of normalizing levels of self-control. However, since the within-group variance decreased after treatment, the frequency of over- and undercontrol was reduced. This tendency was more pronounced in the subsample that achieved possible remission, somewhat contrasting with previous studies of the stability of personality in psychiatric populations [18, 37, 38]. Instead, it aligns with the belief that changes in personality style are, to some extent, a complication of an ED, or a so-called state effect, normalizing with decreasing ED pathology and starvation [10] and previous studies suggesting a change in personality in the direction of healthy controls, induced by recovery [39-41]. However, our exploratory results merely indicate a trend towards normalization of extreme levels of self-control, not a transition from overto undercontrol, which could still be in line with the theory of a stable personality style. In addition, the majority of both the entire group and the subsample achieving possible remission exhibited no reliable changes, and the levels of self-control remained more unchanged than the degree of global anxiety and depression symptoms, which decreased significantly at post-treatment. However, due to the small sample size (n = 14), no conclusions could be drawn from these sub-analyses; the results must instead be considered exploratory.

It might be speculated that patients exhibiting a tendency to normalize levels of self-control during ED treatment or weight restoration, would be unlikely to benefit from interventions based on personality style, assuming that the initial extreme self-control levels were primarily a consequence of starvation, subsiding with weight restoration. However, patients with unchanged selfcontrol levels despite remission or with remaining high levels of ED symptoms could potentially benefit from interventions directed at maladaptive self-control levels. This as both pronounced over- and undercontrol have

Q	EUC-13 pre	EUC-13 post	EUC-13 mean difference	EUC-13 RCI	EUC-13 direction of change	HSCL-25 global pre	HSCL-25 global post	HSCL-25 global mean difference	HSCL-25 global RCl	HSCL-25 global direction of change
1^	1.54	2.54	-	7.4	Towards mean	3.20	2.32	- 0.88	- 4.2	Decrease
2 ^A	2.23	2.08	- 0.15	- 1.1	Towards extremes	3.08	1.64	— 1.44	- 6.9	Decrease
3А	3.46	2.31	- 1.15	- 8.5	Towards mean	3.28	1.72	- 1.56	- 7.5	Decrease
4⊳	2.77	2.23	- 0.54	- 4.0	Towards mean	3.64	2.40	- 1.23	- 5.9	Decrease
5^	2.23	2.08	-0.15	- 1.1	Towards extremes	2.84	2.32	- 0.52	- 2.5	Decrease
6^	2.00	2.15	0.15	1.1	Towards means	2.12	1.24	- 0.88	- 4.2	Decrease
٦A	2.23	2.23	0	0	No change	2.32	1.40	- 0.92	- 4.4	Decrease
8^	2.62	2.46	-0.16	- 1.2	Towards mean	2.24	1.56	- 0.68	- 3.2	Decrease
9^	1.54	2.08	0.54	4.0	Towards mean	1.32	1.24	- 0.08	- 0.4	Decrease
10 ^A	3.38	3.08	- 0.3	- 2.2	Towards mean	2.84	2.12	-0.72	— 3.4	Decrease
11	2.85	2.38	-0.47	- 3.5	Towards mean	3.36	2.92	- 0.44	- 2.1	Decrease
12	1.85	1.54	- 0.31	- 2.3	Towards extremes	3.20	2.24	- 0.96	— 4.6	Decrease
13	2.38	2.31	- 0.07	- 0.5	Towards mean	2.24	1.84	- 0.40	- 1.9	Decrease
14	1.92	1.46	- 0.46	- 3.4	Towards extremes	1.64	2.48	0.84	4	Increase
Self-co table f	ontrol levels are me urther shows calcu	asured with Ego Un lated differences in u	dercontrol scale (EUC mean values, reliable	-13) and symptor change index (RC	ns of anxiety and depression with Hop (1) and direction of change	kins Symptom C	hecklist 25 (HSCL-	25), in 14 participants w	ith an eating disc	order diagnosis. The
2			-							-

Table 3 Individual mean self-control levels and depression/anxiety symptoms, pre- and post-ED treatment (n = 14)

Directions of change for the EUC-13 scores are presented as towards initial sample mean (2.24) or towards the extremes. For the HSCL-25 scores, the direction of change is presented as decrease or increase. Reliable changes are shown in bold

 $^{\rm A}$ Having reached EDE-Q values within one SD of a non-clinical mean (participants 1 – 10)

been shown to correlate with low mental health, potentially increasing the risk for relapse in an ED or another diagnosis frequently associated with maladaptive levels of self-control, such as OCPD [42] or substance abuse [43]. This is hypothetical and not evaluated in this study, and thus requires further research.

Limitations and strengths

This study has several limitations. First, there were several factors influencing levels of anxiety and depression symptoms that we were not able to control for in our analyses, such as traumatic life events, comorbidities, and current medication with for example antidepressants or anxiolytics. Secondly, there are large gender group differences, inhibiting any analyses on gender differences. Furthermore, when controlling for starvation, we used BMI as a proxy, and although BMI can serve as an indication of starvation over time, it is not ideal for assessing current caloric intake.

Additionally, due to the study's cross-sectional design, no conclusions of causality could be drawn from our results. Also, it is important to remember that the EUC-13 cannot distinguish between functional and dysfunctional levels of self-control, but rather measures personality style without considering functional impairments. When dividing the data in categories based on sample mean and standard deviations, the categories become somewhat arbitrary, which may be problematic. Nevertheless, the aim of the study was to examine the highest and lowest levels of over- and undercontrol, with an increased risk of a dysfunctional level, and a division by standard deviation made this possible, making it appear to be an appropriate choice. Further, the outlying groups were defined by the deviation of more than one SD from the mean, which, it may be argued, is not extremely outlying. This division was made to ensure sufficient numbers of individuals in all groups. However, future studies should aim to include an even bigger sample of participants, enabling analyses of more outlying self-control levels.

An obvious limitation to the post-treatment analyses is the small study sample of 14 participants, a result of organizational issues that made data collection difficult. Offering compensation for participation, sending out reminder emails, and offering digital questionnaires could have increased participation rates and should be considered in future studies.

The ecological validity of this study is high, since the study was conducted in a clinical sample. However, patients were excluded when they required inpatient care, at least until their condition stabilized somewhat, which means that the generalizability of the study is limited in those with the most severe illness. The main strength of this study is its number of participants, which was larger than commonly seen when investigating personality style in ED populations. Furthermore, the measures used—EUC-13, HSCL-25 and EDE-Q—are all previously validated in Swedish samples.

Conclusion and future research

Our results indicate that the levels of anxiety/depression symptoms in ED patients are not strongly correlated with levels of self-control, but rather with ED symptoms. Nevertheless, there is a tendency that extreme values of both over- and undercontrol are associated with higher levels of depression, whereas anxiety symptoms seem to increase with increasing levels of undercontrol. Future studies could benefit from considering both overand undercontrol as potentially dysfunctional. In the exploratory analyses self-control levels seemed to move towards normalization after ED treatment for a subset of participants, though levels of anxiety and depression symptoms decreased to a higher degree. Future research should aim to assess the stability of self-control levels post-ED treatment by studying a larger sample. Future studies should also aim to more closely examine potential differences between over- and undercontrolled patients, as well as differentiating between functional and dysfunctional under- and overcontrol and potential gender group differences. A broadened understanding of this field could aid development of more individualized and efficient ED treatments.

Abbreviations

AAN	Atypical anorexia nervosa
AN	Anorexia nervosa
ANOVA	Analysis of variance
ARFID	Avoidant/restrictive food intake disorder
ASD	Autism spectrum disorder
BED	Binge eating disorder
BN	Bulimia nervosa
BMI	Body mass index
ED	Eating disorder
EDE-I	Eating disorder examination interview
EDE-Q	Eating disorder examination questionnaire
DSM-5	Diagnostic and statistical manual of mental disorders, 5th edition
DSM-IV	Diagnostic and statistical manual of mental disorders, 4th edition
EUC-13	Ego undercontrol scale 13
HSCL-25	Hopkins symptom checklist 25
LOESS	Locally estimated scatterplot smoothing
OSFED	Other specified eating disorders
OCPD	Obsessive-compulsive personality disorder
RCI	Reliable change index
SD	Standard deviation

Acknowledgements

We would like to thank all the participants for their involvement in the study. We also want to thank Region Uppsala for making the research possible through clinical research funding.

Author contributions

All authors contributed substantially to each part of the manuscript. The first draft of the manuscript was written by EE, in close collaboration with MI. Statistical analyses were primarily calculated by EE and MI. All authors commented on draft versions of the manuscript. All authors read and approved the final manuscript.

Funding

Open access funding provided by Uppsala University. This research was funded by Region Uppsala who provided a salary to the last author.

Availability of data and materials

Data will not be made publicly available due to confidentiality but can be provided upon reasonable request to the corresponding author.

Declarations

Ethics approval and consent to participate

This study was performed in line with the principles of the Declaration of Helsinki. The study was approved by the Regional Ethics Committee in Uppsala (Ref. nos. 2013/156 and 2014/252).

Consent for publication

Patients signed informed consent regarding publication of their data.

Informed consent

Informed consent was obtained from all individual participants included in the study.

Competing interests

The authors declare that they have no conflict of interest.

Author details

¹Department of Medical Sciences, Child and Adolescent Psychiatry, Uppsala University, Entrance 10, Floor 3B, 751 85 Uppsala, Sweden. ²Department of Clinical Neuroscience, Centre for Psychiatry Research, Karolinska Institutet and Stockholm Health Care Services, Region Stockholm, Solna, Sweden.

Received: 31 October 2022 Accepted: 6 February 2023 Published online: 14 February 2023

References

- American Psychiatric Association. Diagnostic and statistical manual of mental disorders (DSM-5[®]). Washington: American Psychiatric Publishing; 2013. p. 329–54.
- Swinbourne J, Hunt C, Abbott M, Russell J, Clare T, Touyz S. The comorbidity between eating disorders and anxiety disorders: prevalence in an eating disorder sample and anxiety disorder sample. Aust N Z J Psychiatry. 2012;1(46):118–31.
- Casper R. Depression and eating disorders. Depress Anxiety. 1998;1(8):96–104.
- 4. Fennig S, Hadas A. Suicidal behavior and depression in adolescents with eating disorders. Nord J Psychiatry. 2010;64(1):32–9.
- Plana-Ripoll O, Pedersen CD, Agerbo E, Holtz Y, Erlangsen A, Canudas-Romo V, et al. A comprehensive analysis of mortality-related health metrics associated with mental disorders: a nationwide, register-based cohort study. The Lancet. 2019;394(10211):1827–35.
- Södersten P, Bergh C, Leon M, Brodin U, Zandian M. Cognitive behavior therapy for eating disorders versus normalization of eating behavior. Physiol Behav. 2017;15(174):178–90.
- Martinez MA, Craighead LW. Toward person(ality)-centered treatment: how consideration of personality and individual differences in anorexia nervosa may improve treatment outcome. Clin Psychol Sci Pract. 2015;22(3):296–314.
- Turner BJ, Claes L, Wilderjans TF, Pauwels E, Dierckx E, Chapman AL, et al. Personality profiles in eating disorders: further evidence of the clinical utility of examining subtypes based on temperament. Psychiatry Res. 2014;219(1):157–65.

- Westen D, Harnden-Fischer J. Personality profiles in eating disorders: rethinking the distinction between axis I and axis II. Am J Psychiatry. 2001;158(4):547–62.
- Wildes JE, Marcus MD, Crosby RD, Ringham RM, Dapelo MM, Gaskill JA, et al. The clinical utility of personality subtypes in patients with anorexia nervosa. J Consult Clin Psychol. 2011;79(5):665–74.
- 11. Letzring TD, et al. Ego-control and ego-resiliency: generalization of self-report scales based on personality descriptions from acquaintances, clinicians, and the self. J Res Pers. 2005;39(4):395–422.
- 12. Råstam M, Gillberg C, Wentz E. Outcome of teenage-onset anorexia nervosa in a Swedish community-based sample. Eur Child Adolesc Psychiatry. 2003;12(1):i78-90.
- Bohane L, Maguire N, Richardson T. Resilients, overcontrollers and undercontrollers: a systematic review of the utility of a personality typology method in understanding adult mental health problems. Clin Psychol Rev. 2017;1(57):75–92.
- Baudinet J, Simic M, Griffiths H, Donnelly C, Stewart C, Goddard E. Targeting maladaptive overcontrol with radically open dialectical behaviour therapy in a day programme for adolescents with restrictive eating disorders: an uncontrolled case series. J Eat Disord. 2020;8(1):1–13.
- Huey SJ Jr, Weisz JR. Ego control, ego resiliency, and the five-factor model as predictors of behavioral and emotional problems in clinic-referred children and adolescents. J Abnorm Psychol. 1997;106(3):404–15.
- Swendsen JD, Merikangas KR, Canino GJ, Kessler RC, Rubio-Stipec M, Angst J. The comorbidity of alcoholism with anxiety and depressive disorders in four geographic communities. Compr Psychiatry. 1998;39(4):176–84.
- 17. Swendsen JD, Merikangas KR. The comorbidity of depression and substance use disorders. Clin Psychol Rev. 2000;20(2):173–89.
- Kentros M, Smith TE, Hull J, Mckee M, Terkelsen K, Capalbo C. Stability of personality traits in schizophrenia and schizoaffective disorder: a pilot project. J Nerv Ment Dis. 1997;185(9):549–55.
- Krueger RF, Eaton NR. Personality traits and the classification of mental disorders: toward a more complete integration in DSM–5 and an empirical model of psychopathology. Personal Disord Theory Res Treat. 2010;1(2):97–118.
- Mond JM, Hay PJ, Rodgers B, Owen C, Beumont PJV. Validity of the eating disorder examination questionnaire (EDE-Q) in screening for eating disorders in community samples. Behav Res Ther. 2004;42(5):551–67.
- 21. Welch E, Birgegård A, Parling T, Ghaderi A. Eating disorder examination questionnaire and clinical impairment assessment questionnaire: general population and clinical norms for young adult women in Sweden. Behav Res Ther. 2011;49(2):85–91.
- Berg KC, Peterson CB, Frazier P, Crow SJ. Psychometric evaluation of the eating disorder examination and eating disorder examination-questionnaire: a systematic review of the literature. Int J Eat Disord. 2012;45(3):428.
- Isaksson M, Ghaderi A, Wolf-Arehult M, Ramklint M. Overcontrolled, undercontrolled, and resilient personality styles among patients with eating disorders. J Eat Disord. 2021;9(1):47.
- Isaksson M, Ghaderi A, Wolf-Arehult M, Ramklint M. Psychometric properties of the Swedish version of the ego resilience scale (ER) and a new shortened version of the ego undercontrol scale (EUC). Curr Psychol. 2021;40(3):1498–506.
- Vindbjerg E, Mortensen EL, Makransky G, Nielsen T, Carlsson J. A raschbased validity study of the HSCL-25. J Affect Disorde Rep. 2021;4:100096.
- Lundin A, Hallgren M, Forsell Y. The validity of the symptom checklist depression and anxiety subscales: a general population study in Sweden. J Affect Disord. 2015;1(183):247–52.
- Tinghög P, Carstensen J. Cross-cultural equivalence of HSCL-25 and WHO (ten) wellbeing index: findings from a population-based survey of immigrants and non-immigrants in Sweden. Commun Ment Health J. 2010;46(1):65–76.
- Nettelbladt P, Hansson L, Stefansson CG, Borgquist L, Nordström G. Test characteristics of the hopkins symptom check list-25 (HSCL-25) in Sweden, using the present state examination (PSE-9) as a caseness criterion. Soc Psychiatry Psychiatr Epidemiol. 1993;28(3):130–3.
- Peng CYJ, Harwell M, Show-Mann L, Ehman LH. Real data analysis: advances in missing data methods and implications for educational research. Charlotte: IAP; 2006. p. 440.
- Cooper Z, Fairburn C. Eating disorder examination. American Psychological Association; 2011.

- 31. Jager KJ, Zoccali C, Macleod A, Dekker FW. Confounding: what it is and how to deal with it. Kidney Int. 2008;73(3):256–60.
- 32. Asendorpf JB, Denissen JJA, van Aken MAG. Inhibited and aggressive preschool children at 23 years of age: personality and social transitions into adulthood. Dev Psychol. 2008;44(4):997–1011.
- Lynch TR, Hempel RJ, Dunkley C. Radically open-dialectical behavior therapy for disorders of over-control: signaling matters. APT. 2015;69(2):141–62.
- Zucker RA, Heitzeg MM, Nigg JT. Parsing the undercontrol/disinhibition pathway to substance use disorders: a multilevel developmental problem. Child Dev Perspect. 2011;5(4):248–55.
- 35. Díaz-Marsá M, Luis J, Sáiz J. A study of temperament and personality in anorexia and bulimia nervosa. J Pers Disord. 2000;14(4):352–9.
- Heilbrun AB Jr, Bloomfield DL. Cognitive differences between bulimic and anorexic females: self-control deficits in bulimia. Int J Eat Disord. 1986;5(2):209–22.
- Ferguson CJ. A meta-analysis of normal and disordered personality across the life span. J Pers Soc Psychol. 2010;98(4):659–67.
- Boyette LL, Nederlof J, Meijer C, de Boer F, de Haan L. Three year stability of five-factor model personality traits in relation to changes in symptom levels in patients with schizophrenia or related disorders. Psychiatry Res. 2015;229(1):539–44.
- Bloks H, Hoek HW, Callewaert I, van Furth E. Stability of personality traits in patients who received intensive treatment for a severe eating disorder. J Nerv Ment Dis. 2004;192(2):129–38.
- Garner DM, Olmsted MP, Davis R, Rockert W, Goldbloom D, Eagle M. The association between bulimic symptoms and reported psychopathology. Int J Eat Disord. 1990;9(1):1–15.
- 41. Kennedy SH, McVey G, Katz R. Personality disorders in anorexia nervosa and bulimia nervosa. J Psychiatr Res. 1990;24(3):259–69.
- de Reus RJM, Emmelkamp PMG. Obsessive–compulsive personality disorder: a review of current empirical findings. Personal Ment Health. 2012;6(1):1–21.
- Allahverdipour H, Hidarnia AR, Kazemnejad A, Shafii F, Azad Fallah P, Mirzaee E, et al. Assessment of substance abuse behaviors in adolescents': integration of self-control into extended parallel process model. SSU J. 2005;13(1):21–31.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

